Author's response to reviews

Title: Screening and diagnosing depression in women visiting GPs' drop in clinic in primary health care

Authors:

Ranja Stromberg (Ranja.Stromberg@ki.se)
Estera Wernering (Estera.Wernering@hlmserafen.se)
Anna Aberg-Wistedt (Anna.Wistedt@sll.se)
Anna-Karin Furhoff (Anna-Karin.Furhoff@ki.se)
Sven-Erik Johansson (Sven-Erik.Johansson@ki.se)
Lars G Backlund (Lars.G.Backlund@ki.se)

Version: 5 Date: 6 June 2008

Author's response to reviews:

The manuscript is changed only in the format following the instructions. The highlightings are removed and the overstriked text is removed.

The tables are moved to after references. In the figure 1, the term recall visit is changed to repeat visite as this term is used in the manuscript.

In the authors list Lars Backlund's name is corrected to Lars G. Backlund.